Author's response to reviews

Title: Pre-Invasive And Invasive Disease In Women With A Cytological Diagnosis Of High-Grade Lesion And High Grade Lesion Can Not Exclude Microinvasion

Authors:

   Nina S Kuperman (nskuperman@yahoo.com.br)
   Fábio B Russomano (fabiorussomano@gmail.com)
   Yara F Melo (yarafurtadori@terra.com.br)
   Saint Clair S Gomes Jr (scgomes@iff.fiocruz.br)

Version: 8 Date: 20 July 2015

Author's response to reviews: see over
Covering Letter

Dear Ms. April Rada,

Regarding the paper named **Pre-Invasive and Invasive Disease in Women With a Cytological Diagnosis of High-Grade Lesion And High Grade Lesion Can Not Exclude Microinvasion**, we are resubmitting a new version with the issues addressed in your e-mail dated June 30th, as follows.

Regards,

Nina Kuperman.

**Editorial Requirement:**

1. Post-review copyediting: We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at [http://www.biomedcentral.com/authors/authorfaq/editing](http://www.biomedcentral.com/authors/authorfaq/editing).

**Author’s answer:** An English Editor, associated to our institution, has edited our manuscript. If BioMed’s editors still believe the written English needs to be improved, we will send it to Edanz.
2. Acknowledgements: By way of a section ?Acknowledgements?, please acknowledge anyone who contributed towards the article by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for each author, and for the manuscript preparation. Authors must describe the role of the funding body, if any, in design, in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication. Please also acknowledge anyone who contributed materials essential for the study. If a language editor has made significant revision of the manuscript, we recommend that you acknowledge the editor by name, where possible. The role of a scientific (medical) writer must be included in the acknowledgements section, including their source(s) of funding. We suggest wording such as 'We thank Jane Doe who provided medical writing services on behalf of XYZ Pharmaceuticals Ltd.' Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section.

**Author's answer:** Done. Please see line 310.

3. Please include a list of abbreviations that were used on the manuscript. Please include this after the conclusion section of the main article section.

**Author's answer:** Done. Please see line 282.